Author's response to reviews

Title: Screening for inter-hospital differences in cesarean section rates in low-risk deliveries using administrative data. Can it contribute to quality of care initiatives?

Authors:

Willem Aelvoet (willemaelvoet@health.fgov.be)
Francis Windey (francis.windey@health.fgov.be)
Geert Molenberghs (Geert.Molenberghs@uhasselt.be)
Hans Verstraelen (hans.verstraelen@Ugent.be)
Patrick Van Reempts (Patrick.Van.Reempts@uza.be)
Jean-Michel Foidart (jmfoiart@ulg.ac.be)

Version: 2 Date: 30 March 2007

Author's response to reviews:

p 21 I replaced maternal mortality ratio by the maternal mortality rate
I also added the omitted Annexes 1 and 2